Author's response to reviews

Title: Exercise therapy for Stress-related Mental Disorder, a randomised controlled trial in primary care.

Authors:

A. Otto Quartero (a.o.quartero@umcutrecht.nl)
Huib Burger (h.burger@epi.umcg.nl)
Marieke Donker (mariekedonker@zonnet.nl)
Niek J de Wit (n.j.dewit@umcutrecht.nl)

Version: 4 Date: 19 July 2011

Author's response to reviews:

Dear Editorial production team,

Please find herewith the edited version of our paper “Exercise therapy for Stress-related Mental Disorder, a randomised controlled trial in primary care”.

I executed the edits as required. Please let me know if there are any more changes to be made.

Yours sincerely,

A. Otto Quartero